Author's response to reviews

Title: Using Relationship Styles Based on Attachment Theory to Improve Understanding of Specialty Choice in Medicine.

Authors:

Paul S Ciechanowski (pavelcie@u.washington.edu)
Linda LM Worley (worleylindal@uams.edu)
Joan E Russo (jerusso@u.washington.edu)
Wayne J Katon (wkaton@u.washington.edu)

Version: 5 Date: 14 December 2005

Author's response to reviews:

We have made all the requested formatting changes. In addition we have added punctuation, or made consistent brackets throughout the results section for statistical results. In the first paragraph of the Discussion we took the second sentence "This study also demonstrates..." and placed it at the beginning of the second paragraph of the Discussion section so as to improve readability of the paper.